Author's response to reviews

Title: ATBF1 and NQO1 as candidate targets for allelic loss at chromosome arm 16q in breast cancer: Absence of somatic ATBF1 mutations and no role for the C609T NQO1 polymorphism

Authors:

Anne-Marie Cleton-Jansen (a.m.cleton-jansen@lumc.nl)
Ronald Van Eijk (R.van_Eijk@lumc.nl)
Marcel Lombaerts (Lombaerts.m@hsleiden.nl)
Marjanka K. Schmidt (mk.Schmidt@nki.nl)
Laura J. Van ’t Veer (l.vt.veer@nki.nl)
Katja Philippo (K.Philippo@lumc.nl)
Rhyenne M.E. Zimmerman (r.zimmerman@chem.leidenuniv.nl)
Hans Peterse (l.vt.veer@nki.nl)
Vincent T.H.B.M. Smit (V.T.H.B.M.Smit@lumc.nl)
Tom Van Wezel (T.van_Wezel@lumc.nl)
Cees J Cornelisse (C.J.Cornelisse@lumc.nl)

Version: 3 Date: 27 February 2008

Author's response to reviews: see over
Dear editor,

Herewith we send you the revised manuscript nr. 1435322008172029

“ATBF1 and NQO1 as candidate targets for allelic loss at chromosome arm 16q in breast cancer: Absence of somatic ATBF1 mutations and no role for the C609T NQO1 polymorphism”

The manuscript was edited by American Journal Experts (http://www.journalexperts.com), according to their Premium Document Editing protocol where two editors are checking and editing the entire manuscript. They adjusted many grammatical errors and inconsistencies, which have all been checked by me and most times changed in the manuscript according to the instructions of the American Journal Experts team.

We hope that BMC Cancer will now find this manuscript suitable for publication.

Kind regards,

Anne-Marie Cleton-Jansen